Author's response to reviews

Title: A new family with hereditary lysozyme amyloidosis with gastritis and inflammatory bowel disease as prevailing symptoms.

Authors:

estelle jean (estelle.jean@ap-hm.fr)
mikael ebbo (mikael.ebbo@ap-hm.fr)
sophie valleix (sophie.valleix@cch.aphp.fr)
lucas benarous (lucas.benarous@ap-hm.fr)
laurent hey ries (laurent.heyries@ap-hm.fr)
aurelie grados (aurelie.grados@ap-hm.fr)
emmanuelle bernit (emmanuelle.bernit@ap-hm.fr)
gilles grateau (gilles.grateau@tnn.aphp.fr)
thomas papo (thomas.papo@bch.aphp.fr)
brigitte granel (brigitte.granel@ap-hm.fr)
laurent daniel (laurent.daniel@ap-hm.fr)
jean-robert harle (jean-robert.harle@ap-hm.fr)
nicolas schleinitz (nicolas.schleinitz@ap-hm.fr)

Version: 4  Date: 14 February 2014

Author’s response to reviews:

dear editor,

the methods section has been further modified and clarified according to your question in the new version of the manuscript:

- genetic analysis was part of standard care, this has been added
- We added also the sentence: "This retrospective and descriptive study required no additional investigation to standard care and therefore did not require a specific statement of our ethical committee." in the method section.

We hope that these clarification will allow the manuscript to be reviewed

With our best

Nicolas Schleinitz